Author's response to reviews

Title: Patients satisfaction on tuberculosis treatment service and adherence to treatment in public health facilities of sidama zone, south Ethiopia.

Authors:

zekariyas sahile Nezenega (zsahile@yahoo.com)
Yohannes H/Michael Gatho (yohannes.h.michael@gmail.com)
Tadese Ejigu Tafere (tade_et@yahoo.com)

Version: 2 Date: 12 July 2012

Author's response to reviews: see over
Dear sir/Madam I am Zekariyas Sahile, the coauthors of this research. I and my colleagues were conducted this research in the title of “Patients satisfaction on Tuberculosis treatment service and adherence to treatment in public health facilities of Sidama zone, South Ethiopia.”

In most literature states that satisfied patients are the assets of health institution, which will induce a sense of belonging and could improve autonomous participation on health facility development. Thus In order to improve the quality of service as part of total quality management, the measurement of patient satisfaction and adherence to treatment should be assessed.

In this research finding has importance information for administrators and service providers that help to promote patient compliance to treatment by addressing the major predictors of patient satisfaction and adherence to treatment. In addition to that, the data may also prove useful for program planner, evaluator and policy maker to consider factors. I want declare that we have no competing interests hope you will see and feedback your response immediately.

Best regards

Zekariyas Sahile